Author’s response to reviews

Title: Vasorelaxant effects of Angelica decursiva root on isolated rat aortic rings

Authors:

Bumjung Kim (ori-pharm@hanmail.net)
Youngki Kwon (youngki0707@hanmail.net)
Somin Lee (sominlee@naver.com)
Kyungjin Lee (dostudy@naver.com)
Inhye Ham (iham@khu.ac.kr)
Ho Young Choi (hychoi@khu.ac.kr)

Version: 2 Date: 22 May 2017

Author’s response to reviews:

We carried out the experiment by HPLC analysis. And, we corrected the manuscript with HPLC analysis data.